Author's response to reviews

Title: Acute oral toxicity of Insampaedok-san, a traditional herbal formula, in rats and its protective effects against ovalbumin-induced asthma via anti-inflammatory and antioxidant properties

Authors:

Ye-ji Kim (8282tongi@daum.net)
Mee-Young Lee (cozy11@kiom.re.kr)
Woo-young Jeon (ssamggun85@kiom.re.kr)
Ohn-Soon Kim (cozy37@hotmail.com)
Hyeun-kyoo Shin (rain1740@paran.com)

Version: 3 Date: 7 January 2014

Author’s response to reviews:

1. While we thank you for including an ethics statement in your manuscript, at present we do not feel that this is specific enough. We would therefore ask you to confirm whether the protocols of your study were submitted to, and approved by, your institutional ethics committee. Please amend your ethics statement so that this is clearly declared.

-> We appreciate your comment. We added these statements in the Methods section.

• Please include an Authors’ Contributions section in your manuscript. Details on the information that we would ask you to include in this section, as well as formatting guidelines, can be found through the following link:
  http://www.biomedcentral.com/bmccomplementalternmed/authors/instructions/researcharticle#formatting-contributions

-> We appreciate your comment. We added these statements in the end of manuscript and formatted the manuscript according to the guidelines.